Author’s response to reviews

Title: Mothers’ Health Care Seeking Behaviour and Associated Factors for Common Childhood Illnesses, Northwest Ethiopia: Community Based Cross-sectional Study

Authors:
Muluye Molla simeneh (mulumola6@gmail.com)
Mezgebu Yitayal Mengistu (mezgebuy@gmail.com)
Abebaw Addis Gelagay (abebaw.addis@gmail.com)
Mulugeta Tesfa Gebeyehu (mulutesfa.g@gmail.com)

Version: 1 Date: 10 Oct 2017

Author’s response to reviews:

We were commented to look at language with its grammar and we sent for English language fluent speakers who are Teaching English as a Foreign Language for post graduate students and commented strictly.

We were also commented to state about the ethics statement received from ethics committee for verbal consent. We received ethical approval to get verbal consent from our study participants. When we took verbal consent from each study participants, the consent was attached with questionnaire which belongs to each participants. In our study there were no study participants whose age was under age of 16 years. In the previous socio-demographic table we attached, there was age category 15-24 years. since we use age classification (categorization) from EDHS. But in our real data there were no study participants whose age was less than 18 years old.